Author's response to reviews

Title: Affinity for risky behaviors following prenatal and early childhood exposure to tetrachloroethylene (PCE)-contaminated drinking water: a retrospective cohort study

Authors:

Ann Aschengrau (aaschen@bu.edu)
Janice M Weinberg (janicew@bu.edu)
Patricia A Janulewicz (paj@bu.edu)
Megan E Romano (romanom@uw.edu)
Lisa G Gallagher (lisa.g.gallagher@alum.bu.edu)
Michael R Winter (mwinter@bu.edu)
Brett R Martin (brmartin@bu.edu)
Veronica M Vieira (vmv@bu.edu)
Thomas F Webster (twebster@bu.edu)
Roberta F White (rwhite@bu.edu)
David M Ozonoff (dozonoff@bu.edu)

Version: 3 Date: 26 August 2011

Author's response to reviews:

Dear Dr. Granjean,

I inserted a new Table (#4) to the main part of the manuscript that summarizes the key results. I also added some new text to describe this addition. The new material is highlighted in yellow.

Ann Aschengrau